Development of a Guidance Model for the Selection of Organisational Improvement Initiatives

A thesis presented in partial fulfilment of the requirements for the degree of Doctor of Philosophy in Engineering and Industrial Management at Massey University, Palmerston North, New Zealand

> Musli Bin Mohammad 2012

Abstract

This research aims to develop a guidance model for selecting organisational improvement initiatives. The final guidance model was developed using the acronym 'GUIDE' which represents the five key steps to select improvement initiatives: (1) Goal setting, (2) Understanding relevant improvement initiatives, (3) Identifying decision criteria, (4) Deciding on the appropriate initiative, and (5) Evaluating the decision. This research used mixed methods approach, whereby qualitative data was used more dominantly than quantitative data. Two research phases were involved: (1) Development, evaluation and refinement of a conceptual model; and (2) Development, evaluation and refinement of a guidance model. This research incorporated multiphase concurrent and sequential data collection, which comprises an extensive literature review, a document review, a global exploratory survey, an evaluation survey and seventeen semi-structured interviews conducted in New Zealand, Singapore and Malaysia. Semi-structured interviews and an evaluation survey were used as primary sources of data. The proposed multilayer guidance model is one of the first to focus on the holistic processes to be used in selecting improvement initiatives whereby its contents are explicitly aligned to the Business Excellence Models (BEMs), such as Baldrige Criteria for Performance Excellence and European Foundation for Quality Management (EFQM) Excellence Model. Systems and contingency approaches were incorporated in the development of this model so that multiple perspectives and contexts (e.g. current maturity level of an organisation, benefit of implementing the initiative, ability to gain top management commitment and support) are considered when selecting an initiative. Part of the guidance model also consists of a framework that shows the main improvement initiatives that can be adopted towards business excellence (BE), which can help organisations to choose appropriate improvement initiatives by narrowing down the options according to the areas of implementation and BE maturity. This framework also indicates that the BEM can be used as an overarching framework for selecting and managing multiple improvement initiatives. In addition, this research identified that there were 94 national Quality / BE Awards used in 83 countries in year 2010 and revealed one of the latest and comprehensive list of the Quality / BE Awards worldwide.

Table of Contents

Abstracti
Acknowledgementsii
Table of Contentsiii
List of Figuresvii
List of Tablesxi
Abbreviationsxiii
Chapter 1 : Introduction to the research1
1.1 Introduction to the chapter1
1.2 Background to the research1
1.3 Aim and objectives of the research4
1.4 Scope of the research4
1.5 Importance of the research5
1.6 Outline of the thesis7
Chapter 2 : Literature Review 11
2.1 Introduction to the chapter11
2.2 Organisational improvement initiatives11
2.3 Main concepts and/or theories that can be used for the selection of organisational
improvement initiatives 14
2.3.1 Systems approach in selecting and managing organisational improvement
initiatives14
2.3.2 Contingency approach in selecting and managing organisational improvement
initiatives16
2.3.3 Business Excellence Models (BEMs) as a guiding framework for selecting and
managing multiple organisational improvement initiatives
2.3.4 Rational decision making in selecting appropriate improvement initiatives 25
2.4 Prior models on the selection of organisational improvement initiatives
2.5 Conclusion of the chapter
Chapter 3 : Research design and methodology
3.1 Introduction to the chapter
3.2 Selection of research design
3.3 Research procedures
3.4 Data collection methods47
3.4.1 Literature review

3.4.2 Exploratory on-line questionnaire survey	49
3.4.3 Semi-structured interviews	50
3.4.4 Document / record review	51
3.4.5 Evaluation survey	51
3.5 Ensuring quality of research	51
3.5.1 Planning quality	52
3.5.2 Design quality	53
3.5.3 Data quality	54
3.5.4 Interpretive rigour	56
3.5.5 Inference transferability	57
3.6 Ethical considerations	58
3.7 Conclusion of the chapter	58
Chapter 4 : Exploratory Survey – Planning, Implementation and Findings	61
4.1 Introduction to the chapter	61
4.2 Planning and implementation of the exploratory survey	61
4.3 Profiles of survey respondents	64
4.4 Exploratory survey analysis and findings	70
4.5 Conclusion of the chapter	76
Chapter 5 : Interviews – Planning, Implementation and Findings	77
5.1 Introduction to the chapter	77
5.2 Planning and implementation of semi-structured interviews	77
5.3 Profiles of interviewees	81
5.4 Findings from the interviews	86
5.4.1 Reasons for selecting suitable organisational improvement initiatives	87
5.4.2 Current practices of selecting the organisational improvement initiatives	87
5.4.3 People who normally involved in the selection of organisational improvem	ent
initiatives	88
5.4.4 Critical factors that should be considered when selecting suitable	
organisational improvement initiatives	88
5.4.5 Key steps in selecting suitable organisational improvement initiatives	89
5.4.6 Main organisational improvement initiatives that should be used according	g to
the areas of implementation and organisational excellence maturity	90
5.4.7 Criteria, guidelines or models that has been used to assist in the selection	ו of
organisational improvement initiatives	90
5.4.8 Evaluation of several existing models for selecting organisational improve	ment
initiatives	91

5.4.9 Expectations of a guidance model for selecting organisational improve	ement
initiatives	
5.4.10 Evaluation of the conceptual model	
5.4.11 Suggestions to improve the conceptual model	
5.4.12 Evaluation of the refined conceptual model	
5.4.13 Suggestions to improve the refined conceptual model	
5.4.14 Main challenges faced by organisations in selecting suitable organis	ational
improvement initiatives	
5.4.15 Suggested countermeasures to overcome the challenges	
5.4.16 Opinion on whether the selection of suitable organisational improver	nent
initiatives is an important area of study	
5.5 Conclusion of the chapter	
Chapter 6 : Evaluation Survey – Planning, Implementation and Findings	
6.1 Introduction to the chapter	
6.2 Planning and implementation of the evaluation survey	
6.3 Profiles of survey respondents	101
6.4 Evaluation survey analysis and findings	106
6.4.1 Levels of importance of each proposed factor that should be consider	ed when
selecting improvement initiatives	107
6.4.2 Evaluation of the proposed guidance model	109
6.5 Conclusion of the chapter	109
Chapter 7 : Development, Evaluation and Refinement of a Conceptual Model	111
7.1 Introduction to the chapter	111
7.2 Development of a conceptual model	111
7.3 Evaluation of the conceptual model and suggestions for improvement	118
7.3.1 Evaluation of the conceptual model	118
7.3.2 Suggestions for improvement	120
7.4 Refinement of the conceptual model	121
7.5 Evaluation of the refined conceptual model and suggestions for improvem	ent 126
7.5.1 Evaluation of the refined conceptual model	127
7.5.2 Suggestions for improvement	128
7.6 Conclusion of the chapter	131
Chapter 8 : Development, Evaluation and Refinement of a Guidance Model	133
8.1 Introduction to the chapter	133
8.2 Development of a guidance model	133
8.3 Evaluation of the guidance model	143

8.3.1 Degree of agreement for each statement / question related to the GUIDE
model
8.3.2 Strengths, weaknesses and suggestions for improvement
8.4 Refinement of the guidance model 159
8.5 Conclusion of the chapter
Chapter 9 : Discussion and Conclusion
9.1 Introduction to the chapter
9.2 Summary of the main research findings in relation to the research aim and
objectives 181
9.2.1 Main organisational improvement initiatives that should be used according to
the areas of implementation and organisational maturity181
9.2.2 Guidance model currently available to assist in the selection of organisational
improvement initiatives
9.2.3 Main steps to select improvement initiatives
9.2.4 Critical contingency factors to be considered in the selection of improvement
initiatives
9.2.5 Development, evaluation and refinement of an original guidance model for
selecting organisational improvement initiatives
9.3 Contributions of the research
9.3.1 Original contribution to the knowledge
9.3.2 Impact on practice
9.4 Limitations of the research and suggestions for future research
9.5 Conclusion of the chapter
References
Appendices
Appendix 1: List of national Quality / BE Awards worldwide and their adopted BEMs and
administrative organisations 2010A1.1
Appendix 2: Sample of the on-line exploratory surveyA2.1
Appendix 3: Protocol for pilot interviews
Appendix 4: Protocol for first stage interviews
Appendix 5: Protocol for second stage interviews
Appendix 6: Sample of the questionnaire for evaluation surveyA6.1
Appendix 7: Proposed guidance model for selecting organisational improvement
initiativesA7.1
Appendix 8: Final (refined) guidance model for selecting organisational improvement
initiativesA8.1

List of Figures

Figure 1.1: Research focus involving decision making process in selecting improvement
initiatives5
Figure 1.2: Outline of the thesis9
Figure 2.1: An improvement initiative selection framework proposed by
Thawesaengskulthai (2007)
Figure 2.2: Steps involved in evaluating and selecting the initiative (Thawesaengskulthai,
2007)
Figure 2.3: A consultant-based model indicating the examples of initiatives to be used
according to the levels of BE maturity and enablers of Baldrige CPE (Saunders &
Mann, 2007)
Figure 2.4: A simple consultant-based model indicating the examples of initiatives to be
used according to the Baldrige CPE (NZMED, 2006, as cited in NZBEF, 2009) 32
Figure 2.5: A consultant-based model indicating the examples of initiatives to be used
according to the Baldrige CPE (Brown & Pemberton Planning Group Ltd, 2008) 32
Figure 2.6: A guiding model indicating the examples of initiatives to be used according to
the core values and concepts of SQA model (Rimington, 2007)
Figure 2.7: A decision tree for adopting organisational improvement initiatives (Bendell,
2005)
Figure 2.8: A road map for adopting organisational improvement initiatives (Bendell,
2005)
Figure 2.9: A decision tree for adopting organisational improvement initiatives (Radziwill
et al., 2008)
Figure 3.1: Multiphase research design in developing, evaluating and refining conceptual
and guidance model for selecting organisational improvement initiatives
Figure 3.2: Linkages between research aim, research design, research objectives, data
collection methods and sources of data 48
Figure 3.3: Quality domains proposed by O'Cathain (2010) which has been used to
ensure the quality of this research52
Figure 3.4: Iterative processes of developing, evaluating and refining the model for
selecting improvement initiatives54
Figure 3.5: Summary of Chapter 3 and its linkages with other chapters in this thesis 59
Figure 4.1: Response distribution by number of years in which the respondents have
been involved in the areas related to quality, productivity and/or sustainability 65

Figure 4.2: Response distribution by country in which the respondents are currently
working
Figure 4.3: Response distribution by major business activity of the respondents'
organisation67
Figure 4.4: Response distribution by sector of the respondents' organisation
Figure 4.5: Response distribution by approximate number of full-time employees in the
respondents' organisation 69
Figure 4.6: Response distribution by whether the respondents' organisation had won any
BE /quality award(s)
Figure 4.7: Response distribution by whether the respondents' think that selection of
suitable organisational improvement initiatives is an important area of study 70
Figure 5.1: Generating and organising nodes in NVivo 9 80
Figure 5.2: Strategic management process framework (based on document provided by
Interviewee A2)
Figure 6.1: Response distribution by number of years in which the respondents have
been involved in the areas related to quality, productivity and/or organisational
excellence
Figure 6.2: Response distribution by role of the respondents in the organisation 103
Figure 6.3: Response distribution by country in which the respondents are currently
based 104
based
 based
 based
 based
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135 Figure 8.2: Response distribution by whether the respondents' think that the Introduction 140
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135 Figure 8.2: Response distribution by whether the respondents' think that the Introduction Section to the GUIDE Model is written clearly and easy to understand (Total
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135 Figure 8.2: Response distribution by whether the respondents' think that the Introduction Section to the GUIDE Model is written clearly and easy to understand (Total responses = 31) 145
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation. 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards. 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135 Figure 8.2: Response distribution by whether the respondents' think that the Introduction Section to the GUIDE Model is written clearly and easy to understand (Total responses = 31) 145 Figure 8.3: Response distribution by whether the respondents' think that the GUIDE 145
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135 Figure 8.2: Response distribution by whether the respondents' think that the Introduction Section to the GUIDE Model is written clearly and easy to understand (Total responses = 31) 145 Figure 8.3: Response distribution by whether the respondents' think that the GUIDE model is presented clearly and easy to understand (Total responses = 31) 146
based 104 Figure 6.4: Response distribution by major business activity of the respondents' organisation. 105 Figure 6.5: Response distribution by whether the respondents' organisation had won any BE / quality awards. 106 Figure 7.1: A conceptual model for selecting organisational improvement initiatives 113 Figure 7.2: Overview of the refined conceptual model for selecting organisational improvement initiatives 123 Figure 8.1: Overview of the guidance model for selecting organisational improvement initiatives 135 Figure 8.2: Response distribution by whether the respondents' think that the Introduction Section to the GUIDE Model is written clearly and easy to understand (Total responses = 31) 145 Figure 8.3: Response distribution by whether the respondents' think that the GUIDE model is presented clearly and easy to understand (Total responses = 31) 146 Figure 8.4: Response distribution by the level of agreement of the respondents with the 140

Figure 8.5: Response distribution by the level of agreement of the respondents with the steps and content of the 'U - Understand organisational context' section (Total Figure 8.6: Response distribution by the level of agreement of the respondents with the steps and content of the 'I - Identify relevant organisational improvement initiatives' Figure 8.7: Response distribution by the level of agreement of the respondents with the steps and content of the 'D – Decide appropriate initiative' section (Total responses Figure 8.8: Response distribution by the level of agreement of the respondents with the steps and content of the 'E – Evaluate the decision' section (Total responses = 31) Figure 8.9: Response distribution by whether the respondents' think that the model is useful for quality and/or organisational improvement consultants (Total responses = Figure 8.10: Response distribution by whether the respondents' think that the model is useful for quality and/or organisational improvement practitioners (Total responses Figure 8.11: Response distribution by whether the respondents' think that the model is useful for Small and Medium Sized Enterprises (SMEs) (Total responses = 31).. 152 Figure 8.12: Response distribution by whether the respondents' think that the model is Figure 8.13: Response distribution by whether the respondents' think that the model is Figure 8.14: Interface of the refined (final) guidance model for selecting organisational Figure 8.15: A framework showing 30 main initiatives that can be implemented to improve organisational performance according to the common enabling criteria of Figure 9.1: Linkages between research problem / issue, aim, objectives, data collection Figure 9.2: The links between sub-section 9.2.1 and other sections in this thesis 183 Figure 9.3: Filtration processes involved in the identification of the main improvement Figure 9.4: Processes involved in the development of a framework that indicates the main organisational improvement initiatives to be used according to the areas of

Figure 9.5: The links between the research aim, first research objective and research
questions for the first research objective 187
Figure 9.6: The links between Section 9.2.2 and other sections in this thesis 188
Figure 9.7: The links between the research aim, second research objective and research
questions for the second research objective 189
Figure 9.8: A business improvement initiative selection model for consulting firms
(Wieleman, 2011)
Figure 9.9: The links between Section 9.2.3 and other sections in this thesis 193
Figure 9.10: A framework showing several main steps involved in the selection of
improvement initiatives194
Figure 9.11: The links between the research aim, third research objective and research
question for the third research objective 194
Figure 9.12: Processes involved in the development of a framework that indicates the
main steps to select improvement initiatives
Figure 9.13: The links of Section 9.2.4 with other sections in this thesis
Figure 9.14: The links between the research aim, fourth research objective and research
questions for the fourth research objective 198
Figure 9.15: Processes involved in the development of a final GUIDE model for selecting
organisational improvement initiatives 201
Figure 9.16: The links between the research aim, fifth research objective and research
questions for the fifth research objective 202

List of Tables

Table 2.1: Definitions and examples of approach, management system, tool and
technique for improving organisational performance13
Table 2.2: Common criteria of BEMs based on comparison of major BEMs and previous
research
Table 2.3: Descriptions of the common enabler criteria of BEM and their core values and
concepts22
Table 2.4: Comparison of steps involved in making decision
Table 2.5: Comparison of seven prior models that can be used to assist in the selection
of improvement initiatives27
Table 3.1: Summary of research questions together with research objectives, data
collection methods and types of collected data43
Table 3.2: Classification of mixed methods research designs
Table 4.1: Degree of suitability in terms of where the improvement initiative should be
used according to the implementation areas72
Table 4.2: List of organisational improvement initiatives that have a 'very high suitability'
(median = 5) to be implemented according to the implementation areas
Table 5.1: Breakdown of the interviewees according to country, position, category, type
of industry and sector81
Table 5.2: Profiles of the interviewees 82
Table 5.3: Areas of inquiry for each stage of interviews
Table 5.4: Strengths, limitations and/or improvement suggestions of several existing
models for selecting organisational improvement initiatives based on feedback from
interviewees
Table 6.1: List of potential respondents for evaluation survey 100
Table 6.2: Distribution of the respondents and response rate 101
Table 6.3: Survey results on the levels of importance of each proposed factor to be
considered when selecting improvement initiatives108
Table 7.1: Descriptions of the BE maturity levels according to the areas of
implementation
Table 8.1: Proposed Decision / Selection Criteria
Table 8.2: Brief explanations of the levels of BE maturity
Table 8.3: Brief descriptions of the 30 main organisational improvement initiatives widely
used by organisations167
Table 8.4: Proposed criteria for the selection of organisational improvement initiatives171

Abbreviations

AHP	Analytic Hierarchy Process
AIRMIC	Association of Insurance and Risk Managers
ALARM	National Forum for Risk Management in the Public Sector
APO	Asian Productivity Organisation
BE	Business Excellence
BEM	Business Excellence Model
BCM	Business Continuity Management
BI	Business Improvement
BPIR	Business Performance Improvement Resource
BPR	Business Process Reengineering
CEHE	Consortium for Excellence in Higher Education
CEO	Chief Executive Officer
COER	Centre for Organisational Excellence Research
CPE	Criteria for Performance Excellence
EFQM	European Foundation for Quality Management
ERM	Enterprise Risk Management
FMEA	Failure Mode and Effect Analysis
GUIDE	Goal-Understand-Identify-Decide-Evaluate
IRM	Institute of Risk Management
ISO	International Organisation for Standardisation
KM	Knowledge Management
MADM	Multiple Attribute Decision Making
MCDM	Multiple Criteria Decision Making
NIST	National Institute of Standards and Technology
NZAS	New Zealand Aluminium Smelters
NZBEF	New Zealand Business Excellence Foundation
NZMED	New Zealand Ministry of Economic Development
NZOQ	New Zealand Organisation for Quality
PESTLE	Political-Economic-Social-Technological-Legal-Environmental
PDCA	Plan-Do-Check-Act
QC	Quality Control

QFD	Quality Function Deployment
ROI	Return of Investment
SAW	Simple Additive Weighting
SME	Small and Medium Sized Enterprises
SPC	Statistical Process Control
SPRING	Standards, Productivity and Innovation Board Singapore
SQA	Singapore Quality Award
SWOT	Strengths-Weaknesses-Opportunities-Threats
TOPSIS	Technique for Order Preference by Similarity to Ideal Solution
ТРМ	Total Productive Maintenance
TQM	Total Quality Management
WSM	Weighted Sum Model

Chapter 1: Introduction to the research

1.1 Introduction to the chapter

This chapter explains the research background, aim and objectives, scope, and importance. It concludes with a description of the outline of the thesis.

1.2 Background to the research

To remain relevant, competitive and sustainable in today's increasingly complex, uncertain and dynamic environment, organisations are required to improve their performance in order to meet the purpose of their existence, and to satisfy and exceed the expectations of customers, employees, shareholders, supply chain partners, community and other stakeholders (Foley, 2010; Harrington & Harrington, 1995; Porter, 1998; Slack, Chambers, Johnston, & Betts, 2009). Organisations are facing problems in selecting appropriate improvement initiatives due to a plethora of initiatives currently available in the market (English, 1998; Hendra, 2010; Thawesaengskulthai, 2010). 'Improvement initiatives' refers herein to approaches, systems, tools and/or techniques and include, for example: Six Sigma, Lean, Business Process Reengineering, ISO9001, and benchmarking (Van der Wiele, Van Iwaarden, Dale, & Williams, 2007). Even more challenging is that the number of improvement initiatives increases every year, which makes it even harder to select the most appropriate initiative (Baxter & MacLeod, 2008; Davenport, Prusak, & Wilson, 2003; Thawesaengskulthai, 2010).

A number of organisations view initiatives such as these as a potential panacea for all organisational problems (Ricondo & Viles, 2005). In reality, while none of the individual initiatives can solve all problems effectively in the organisation, each initiative has a role to play towards improving organisational performance. Every initiative has its own strengths and limitations (Francis, 2010). Some initiatives are more effective under certain conditions and contexts (Ricondo & Viles, 2005). In process improvement, for instance, Six Sigma is more effective for reducing variation, Lean for eliminating non value-adding processes or activities, and Theory of Constraints for identifying and 'elevating', or dealing with process constraints (Nave, 2002).

In the quality and organisational improvement community, there can be considered to be certain factions, which comprise those who are primarily oriented towards ISO9000, Six Sigma, Lean and others. Each of these groups competes among one another and tends to solve the problem through the lens of a particular initiative, without seeing the complementary nature of the different initiatives (Cobb, 2003). It is argued, however, that application of such initiatives in isolation without proper planning and strategy will only provide short-term benefits (Dale, 2007; Dale & McQuater, 1998; Ricondo & Viles, 2005).

Improvement initiatives swing in and out of fashion similar to clothing style, car design and music trends (Clark, 2004). According to Cobb (2003, p. 10), "Every time a new management technique comes into vogue, whatever came before it is tossed out and forgotten and the new approach becomes a 'paradigm' for redefining how the business is managed." An effect of this phenomenon is that organisations become the market for the latest management fashion, and managers tend to search for new initiatives (Cobb, 2003; Seddon, 2003). In response to this issue, Cobb (2003) asserts that the search for new initiatives is not the absolute answer. It is more crucial that people should have deeper understanding of how the organisation operates or should operate as a system, and carefully select the right initiative for the right situation (Basu & Wright, 2005; Cobb, 2003; Francis, 2010). Slack et al. (2009) pointed out that:

The problem lies not with new improvement ideas, but rather with some managers becoming a victim of the process, where some new idea will entirely displace whatever went before. Most new ideas have something to say, but jumping from one fad to another will not only generate a backlash against any new idea, but also destroy the ability to accumulate the experience that comes from experimenting with each one. (p.451)

In other cases, many organisations have failed to reap the benefit of implementing the improvement initiatives. One of the reasons why this happens is due to lack of clear understanding by people regarding when, where and how to implement the initiatives (Kwok & Tummala, 1998). Many people have also not considered the contextual factors when selecting and implementing improvement initiatives at their

organisation. The right initiatives to be used may vary depending on several contextual factors, such as: the current maturity level of an organisation, areas in which the initiatives are adopted, type or size of an organisation and the capabilities of its workforce (Benson, Saraph, & Schroeder, 1991; Dahlgaard & Dahlgaard-Park, 2004; National Institute of Standards and Technology [NIST], 2010).

Most of the previous studies only focused on one specific initiative, such as, benchmarking (Adebanjo & Mann, 2008a; Mann & Grigg, 2004; McAdam & Kelly, 2002), ISO9000 (Bendell, 2000; Casadesus & Karapetrovic, 2005; Van der Wiele, Williams, & Dale, 2000), and Six Sigma (Antony, 2007; Antony & Banuelas, 2002; Basu, 2004a). Each of these studies tends to promote the particular initiative and goes into detail about the purpose, strengths, limitations and/or implementation process of the initiative. On the other hand, there are also several surveys undertaken to identify the trends, usage and/or effectiveness of improvement initiatives, which includes Cullen, O'Connor, and Mangan (2004), Mann (2008a), Rigby and Bilodeau (2007) and Weiler (2004). Unfortunately, only a few studies been found (such as, Bendell, 2005; Radziwill, et al., have 2008; Thawesaengskulthai, 2007) to address how to manage multiple improvement initiatives and provide guidance on how to select appropriate initiatives. As a result, there are limited number of models and/or guidelines currently available to assist organisations in selecting and managing multiple improvement initiatives. Most of the existing models do not explicitly stress the importance of understanding organisational profiles and improvement initiatives before selecting an appropriate initiative. In addition, none of the existing guidance models have explicitly aligned the selection processes with the Business Excellence Models (BEMs) although there is a demand for this alignment from the users and administrators of BEMs.

It is also evident from literature research (such as, Francis, 2010; Radziwill, et al., 2008; Thawesaengskulthai, 2010; Thawesaengskulthai & Tannock, 2008a) and from discussions with quality experts and practitioners that many organisations need guidance on what initiatives to use, in what order to implement the initiatives and how to select suitable initiatives. To address this problem, this research was conducted to develop meaningful guidelines for selecting appropriate improvement initiatives according to the context.

1.3 Aim and objectives of the research

This research aims to develop a guidance model for selecting appropriate organisational improvement initiatives. In order to achieve the research aim, the following objectives were set:

- To identify the main organisational improvement initiatives that should be used according to the areas of implementation and organisational maturity;
- (ii) To identify the guidance models currently available to assist in the selection of organisational improvement initiatives;
- (iii) To investigate the main steps involved in selecting organisational improvement initiatives;
- (iv) To identify the critical contingency factors that should be considered in selecting organisational improvement initiatives; and
- To develop, evaluate and refine an original guidance model for selecting appropriate organisational improvement initiatives.

1.4 Scope of the research

The scope of the research is described as follows:

- (i) This research focuses on the decision making process in selecting appropriate organisational improvement initiative, which is depicted in Figure 1.1. It does not cover the adoption and maintenance of initiatives.
- (ii) This research is intended for practitioners, managers, consultants, researchers, and/or academics in the area of quality and business excellence (BE).
- (iii) This research focuses on the usage of rational and structured approach in selecting improvement initiatives.



Figure 1.1: Research focus involving decision making process in selecting improvement initiatives

1.5 Importance of the research

Selection of improvement initiatives is an important area of research due to the following main reasons:

- Previous literature have highlighted the importance of selecting the right initiative for a given context or situation, such as Basu (2004b), Francis (2010), Hendra (2010) and Rigby and Bilodeau (2005). Selection processes will help organisations doing the right thing. As stated by Ackoff (1999, p. 10), "It is better to do the right thing wrong than to do wrong thing right. When we do the right thing wrong, we make mistake that can be corrected; hence we learn how to be more effective". Therefore, selection and adoption of the most appropriate improvement initiative according to the situation are really crucial to the organisations.
- The adoption of initiatives requires time, resources, financial and knowledge (Thawesaengskulthai, 2007). To avoid unnecessary waste and frustration, it

would be better for people to select the right initiative that will fit with organisation's context and provide value to the organisation.

- Many organisations need guidance and advice on the selection of appropriate improvement initiatives due to a myriad of initiatives currently available and the number of initiatives is increasing every year (Thawesaengskulthai, 2007).
- Selection, management and/or integration of multiple improvement initiatives is an important quality issue in the future and become one of the main competencies required for future quality professional (Burnell, 2008).
- There is a relatively very few academic publications and only one PhD thesis (Thawesaengskulthai, 2007) have been found focussing on the selection of improvement initiatives. Therefore, this research is important to enrich the pool of reference materials and findings relating to this important subject matter.

A global on-line exploratory survey was conducted to investigate the importance of the research. The respondents of the survey were the practitioners, managers, consultants, and academicians who have good understanding and experience on improvement initiatives. Further details of this survey are discussed in Chapter 4. Fifty-nine (59) respondents answered the question related to this issue and all of them, 100%, agreed that the selection of suitable improvement initiatives is an important area of study. Thirty respondents (51% of total responses) indicated that the study is 'extremely important', 23 respondents (39%) indicated 'high importance' and the balance of six respondents (10%) indicated 'moderate importance'. All the interviewees also agreed that this study is important. This research also makes an original contribution to the body of knowledge and gives impact on practice, which will be discussed in Chapter 9.

1.6 Outline of the thesis

This thesis consists of nine chapters as depicted in Figure 1.2. The first chapter introduces the background, aim, objectives, scope, and importance of the research. Chapter 2 elaborates the critical review of literature related to the organisational improvement initiatives, main concepts and/or theories that can be used for the selection of improvement initiatives (Systems theory, Contingency theory, BE and rational decision making), and previous models on the selection of improvement initiatives to the research as well as identifies gap in current knowledge of selecting improvement initiatives.

Chapter 3 describes the research design and methodology. This chapter consists of the selection of research design, research procedures, data collection methods, ensuring quality of research and ethical considerations. All the data collection methods (literature review, exploratory survey, semi-structured interviews, document review and evaluation survey) are briefly explained in this chapter. A detailed explanation on the planning, implementation and findings of the exploratory survey, interviews, and evaluation survey is provided in Chapter 4, 5 and 6. This study uses multiphase research design which comprises two research phases: (1) Development, evaluation and refinement of a conceptual model and (2) Development, evaluation and refinement of a guidance model. These two research phases are further discussed in Chapter 7 and 8.

Chapter 4 explains about the exploratory survey. It consists of the planning and implementation of the exploratory survey, profiles of survey respondents, and exploratory survey analysis and findings.

7

Chapter 5 describes about the semi-structured interviews. It comprises of the planning and implementation of the interviews, profiles of interviewees, and findings from the interviews. Results from the document review are indirectly explained in this chapter.

Chapter 6 explains about the evaluation survey. It consists of the planning and implementation of the evaluation survey, profiles of survey respondents, and evaluation survey analysis and findings.

Chapter 7 describes the first phase of research involving the development, evaluation and refinement of a conceptual model for selecting improvement initiatives. It comprises: (1) development of a conceptual model, (2) evaluation of the conceptual model and suggestions for improvement, (3) refinement of the conceptual model, and (4) evaluation of the refined conceptual model and suggestions for improvement.

Chapter 8 describes the second phase of research involving the development, evaluation and refinement of a guidance model for selecting improvement initiatives. It consists of: (1) development of a guidance model, (2) evaluation of the guidance model, and (3) refinement of the guidance model.

Chapter 9 concludes the main findings in relation to the research aim and objectives, explains the contributions of the research and describes the limitations of the research and suggestions for future research. This chapter also explains the linkages between research problem / issue, aim, objectives, data collection methods and sources of data (for example, see Figure 9.1).

8



Figure 1.2: Outline of the thesis

Chapter 2: Literature Review

2.1 Introduction to the chapter

This chapter reviews the relevant literature related to the selection of organisational improvement initiatives. First of all, the term 'organisational improvement initiatives' is explained. Then, the four main concepts and/or theories that can be used in the selection of improvement initiatives are described: (1) contingency theory, (2) systems theory, (3) Business Excellence Models (BEMs), and (4) rational decision making. It is followed by explanation of previous improvement initiatives selection models. Finally, a conclusion for this chapter is presented.

2.2 Organisational improvement initiatives

Performance improvement, change and innovation are part and parcel of organisational life (Baxter & MacLeod, 2008). Organisations need to improve their performance in order to meet the purpose of their existence, and to satisfy and exceed the expectations of customers, employees, shareholders, supply chain partners, community and other stakeholders (Foley, 2010). Organisations should endeavour to improve faster than their competitors to stay ahead in their market segment.

In general, organisational improvement can be categorised into continuous improvement and breakthrough improvement (Imai, 1986; Slack et al., 2009). 'Continuous improvement', also known as 'Kaizen', is a never-ending, small and incremental performance improvement involving everyone in the organisation (Imai, 1986, 1997; Slack et al., 2009). It is normally people oriented, based on common sense and use a relatively low-cost approach (Imai, 1986, 1997). In contrast, the 'breakthrough' or 'innovation-based' improvement is a major and dramatic non-incremental performance improvement based on technological breakthrough and/or new inventions (Imai, 1986, 1997; Slack et al., 2009). According to Slack et al. (2009, p. 439), this type of improvement "can be expensive, often disrupting the ongoing workings of the operation, and frequently involving changes in the product/service or process technology". To survive in the 21st century, both types of

improvement should be adopted by the organisations in order to achieve performance excellence (Harrington, 1995; Thawesaengskulthai, 2007). In relation to this, Hayes et al. (2005) stated that:

The great risk of the incremental approach is being leapfrogged...by a competitor that abandons its traditional technology, location, or corporate strategy and adopts a new and more successful one...Conversely, the great risk of strategic leap approach is that a new breakthrough may not be available exactly when it is needed...An obvious response in such eventuality is for the company to adopt an incremental approach until a breakthrough does become possible. (p.286)

Harrington and Lomax (2000) and the Business Performance Improvement Resource website (www.bpir.com) have listed more than one thousand improvement initiatives that can be used by organisations. In addition to this, Cameron and Barnett (1999, p. 286) highlighted that: "the American Quality Foundation's (1992) survey of companies initiating quality improvement programs found that more than 945 quality tactics, tools, and techniques had been employed". It is also expected that the number of improvement initiatives will continue to increase every year concurrent with the increment of books and publications connected to performance and/or organisational improvement (Baxter & MacLeod, 2008; Davenport et al., 2003).

Organisational improvement initiatives are also known as management tools (Rigby 2005), & Bilodeau, quality management and improvement initiatives (Thawesaengskulthai, 2007), business process improvement methodologies (Bendell, 2005) and performance improvement methods (Harrington & Lomax, 2000). These improvement initiatives were mostly developed by the management gurus, consultants, academics and/or practitioners (Baxter & MacLeod, 2008; Davenport et al., 2003; Greatbatch & Clark, 2005). Improvement initiatives in the present context refer to approaches, management systems, tools and/or techniques (Van der Wiele et al., 2007). Definitions of an approach, management system, tool and technique together with some examples of initiatives are provided in Table 2.1. Each improvement approach (e.g. Six Sigma, Lean, Total Quality Management [TQM], Business Process Reengineering [BPR]) has its own set of management systems, tools and/or techniques. For example, the following tools and techniques are widely used in the adoption of Six Sigma: Statistical Process Control (SPC),

Failure Mode and Effect Analysis (FMEA), Quality Function Deployment (QFD), and Control Chart (Ricondo & Viles, 2005). As for TQM, the following management systems, tools and techniques may be used: ISO9000, Control Chart, Cause and Effect Diagram, QFD, and Benchmarking (Hellsten & Klefsjö, 2000).

Table 2.1: Definitions and examples of approach, management system, too	and
technique for improving organisational performance	

No	Item	Definition	Examples
1	Approach	An approach needs resources (e.g. training, hiring additional and specific personnel), senior management commitment, strategic planning and an "intellectual effort in term of its deployment and adoption" (Van der Wiele et al., 2007, p. 561).	TQM, BPR, Six Sigma, Lean
2	Management system	"A system comprises written information in the form of instructions and procedures in order to direct and control some form of operation" (Van der Wiele et al., 2007, p. 561)	Quality Management System (ISO9000), Environmental Management System (ISO14000), Occupational Health and Safety Management System
3	Tool	A tool can be "described as a device which has a clear role and defined application. It is often narrow in its focus and can be and is usually used on its own" (Dale, 1993, as cited in Van der Wiele et al., 2007, p. 562)	Cause and Effect Diagram, Pareto Diagram, Control Chart, Histogram, Relationship diagram, Flowchart
4	Technique	A technique "has a wider application than a tool". It requires "more thought, skill, knowledge, understanding and training to use them effectively. A technique may even be viewed as a collection of tools" (Dale, 1993, as cited in Van der Wiele et al., 2007, p. 562)	SPC, Benchmarking, QFD, FMEA

These improvement initiatives can be linked, aligned and/or integrated in order to eliminate redundancies and improve system efficiency and effectiveness. Six Sigma is widely integrated with Lean and known as Lean Six Sigma (Arnheiter & Maleyeff, 2005; Byrne, Lubowe, & Blitz, 2007). Ricondo and Viles (2005) explains how Six Sigma can be linked to TQM, BPR, Lean and the Learning Organisation. Numerous literature also discusses about the integration of Quality Management System (e.g. ISO9001), Environmental Management System (e.g. ISO14001) and/or Occupational Health and Safety Management System (e.g. OHSAS 18001), including Abarca (1998), Douglas and Glen (2000), Karapetrovic and Willborn (1998), Mohammad, Osman, Yusuff, and Ismail (2005) and Zutshi and Sohal (2005).

Slack, et. al. (2009) and Slack and Lewis (2008) summarise several expected benefits of adopting these improvement initiatives which include but are not limited to:

- Improve quality (e.g. reduce defects per units, reduce level of customer complaints).
- Improve dependability (e.g. reduce percentage of orders delivered late, better schedule adherence).
- Improve speed (e.g. improve order lead time, improve cycle time).
- Better flexibility (e.g. reduce time needed to develop new products / services, better range of products / services).
- Reduce cost (e.g. reduce cost per operation hour, better utilisation of resources, reduce transaction cost, improve labour productivity).

2.3 Main concepts and/or theories that can be used for the selection of organisational improvement initiatives

This section explains four main concepts and/or theories that can be used for the selection of organisational improvement initiatives: (1) Systems theory, (2) Contingency theory, (3) BE, and (4) Rational decision making.

2.3.1 Systems approach in selecting and managing organisational improvement initiatives

Kast and Rosenzweig (1985, p. 15) define a system as "an organized, unitary whole composed of two or more interdependent parts, components, or subsystems and delineated by identifiable boundaries from its environmental suprasystem". A system can also be defined as "a complex whole the functioning of which depends on its parts and the interactions between those parts" (Jackson, 2003, p. 3).

Systems approach helps people to be aware about the "interrelationships among sub-systems" and "interactions between system and its suprasystem" (Kast & Rosenzweig, 1985, p. 15). According to Ackoff (1999):

A system is a whole that cannot be divided into independent parts without loss of its essential properties or functions...For example, no part of an automobile by itself can transport a person from one place to another, nor can any part of a person live when separated from him or her. (p.8) In addition to this, Ackoff (1999) also highlighted that:

When the performances of the parts of a system, considered separately, are improved, the performance of the whole may not be (and usually is not) improved...The performance of a system depends on how its parts interact, not on how they act taken separately. (p.9)

Therefore, it is crucial to use systems and holistic approach in managing organisational improvement as opposed to the piecemeal approach (Kast & Rosenzweig, 1985). With regards to the selection of improvement initiatives, it is also important to consider whether the initiatives to be used are compatible and can interact well with the existing and future organisation system (Dale, 2007).

Subsequently, it is also important to understand feedback mechanism in a system. Kast and Rosenzweig (1972) stated that:

Information concerning the outputs or the process of the system is fed back as an input into the system, perhaps leading to changes in the transformation process and/or future outputs. Feedback can be both positive and negative...Negative feedback is informational input which indicates that the system is deviating from a prescribed course and should readjust to a new steady state. (p. 450)

In general, systems can be categorised as closed and open system. According to Kast and Rosenzweig (1972):

Open systems exchange information, energy, or material with their environments. Biological and social systems are inherently open systems; mechanical systems may be open or closed...The open system can be viewed as a transformation model. In a dynamic relationship with its environment, it receives various inputs, transforms these inputs in some way, and exports outputs...systems have boundaries which separate them from their environments. The concept of boundaries helps us understand the distinction between open and closed systems. The relatively closed system has rigid, impenetrable boundaries; whereas the open system has permeable boundaries between itself and a broader suprasystem. (p.450) Modern organisation theories consider the organisation as an open system as opposed to the traditional organisation theories that view the organisation as a closed system (Ashmos & Huber, 1987; Kast & Rosenzweig, 1985). Organisation can also be seen as a system that consists of various subsystems and as a subsystem of a larger external environment system (Johnson, Kast, & Rosenzweig, 1964). Examples of the subsystems within the organisation system are quality system, financial and costing system, human resources system, management information system and production system. Meanwhile, external environment systems of an organisation include external political system, economic system, social system, technological system, legal system and environmental system (Karapetrovic & Willborn, 1998; Kast & Rosenzweig, 1985).

By adopting a systems approach in managing organisational improvement, people should be able to understand that the organisation consists of various interlinked processes that convert input (e.g. materials, money, human resources, information) into output (e.g. products, services, stakeholder satisfaction) and interact with its external environment in order to achieve the goal and purpose of its existence. Better output can be obtained by improving the input and processes within the organisation's system (Dean & Bowen, 1994; Kast & Rosenzweig, 1985). The concept of input, process and output will also be incorporated in the development of a guidance model for selecting improvement initiatives.

2.3.2 Contingency approach in selecting and managing organisational improvement initiatives

According to Kast and Rosenzweig (1985):

The contingency view seeks to understand the interrelationships within and among subsystems as well as between the organization and its environment and to define patterns of relationships or configurations of variables... Contingency views are ultimately directed toward suggesting organizational designs and managerial actions most appropriate for specific situations. (p.116) Kast and Rosenzweig (1985) also added that:

The essence of this view is that there is no one best way and that there is a middle ground between 'universal principles' and 'it all depends'. This approach recognize the complexity involved in managing modern organizations but uses the existing body of knowledge to relate environment and design, to match structure and technology, to integrate strategy and tactics, or to determine the appropriate degree of subordinate participation in decision making, given a specific situation. (p.18)

Many existing literature have highlighted the importance of using contingency approach in managing organisation and understanding complex and dynamic situations, for instance Donaldson (2001), Koontz and O'Donnell (1976), Lawrence and Lorsch (1967), and Sousa and Voss (2008). In relation to this, several examples of main contingency factors that should be considered in selecting appropriate initiatives have been included in the guidance model. Since every organisation is unique, there is no one best initiative that can solve all organisational problems. Each initiative also has its own purpose, strengths and limitations. The most appropriate initiative depends on the context in which it is adopted, rather than assumed to be universally applicable. Managers and/or leaders in the organisations should therefore consider all related contingency factors before selecting the right initiative for the situation. The examples of contingency factors include but are not limited to:

- Capability of the workforce to implement the improvement initiatives (NIST, 2010; Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a).
- Organisational culture (Sousa & Voss, 2008; Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a).
- Top management commitment and support (Benson et al., 1991; Saunders & Mann, 2007; Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a).
- Expected costs, time and resources needed to introduce and implement the initiatives successfully (Dale, 2007).
- Vision and mission of the organisation (Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a).
- Direction, strategic plan and goals of the organisation (Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a).

- External environment in which the organisations operate, which includes: political, economic, social, technological, legal and environmental factors (Capon, 2004).
- Level of organisational excellence maturity (Dahlgaard & Dahlgaard-Park, 2004; NIST, 2010).
- Ability of the initiative to "fit in with, complement", integrate, and/or "support" other initiatives "already in place, and might be [used] in the future" (Dale, 2007, p.338).
- Types (e.g. private, public or non-profit) and sizes (e.g. small, medium or large) of the organisation (Benson, et al., 1991; Dahlgaard & Dahlgaard-Park, 2004; NIST, 2010; Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a).
- Expected value / benefit of using the initiative (Thawesaengskulthai, 2007; Thawesaengskulthai & Tannock, 2008a, 2008b).
- Areas in which the initiatives will be used (Saunders & Mann, 2007).

2.3.3 Business Excellence Models (BEMs) as a guiding framework for selecting and managing multiple organisational improvement initiatives¹

There are many definitions of BE and it varies depending on the areas of study. Several definitions of BE based on Quality Management area are listed as follows:

- "Excellence in strategies, business practices, and stakeholder²-related performance results that have been validated by assessments using proven business excellence models" (Adebanjo & Mann, 2008b, p. 1).
- "Overall way of working that balances stakeholders concerns and increases the probability of long-term organisational success through operational, customer-related, financial, and marketplace performance excellence"

 ¹ A large part of this section was published in the International Journal of Total Quality Management and Business Excellence (see Mohammad, Mann, Grigg, & Wagner, 2011a).
 ² Stakeholder is defined as "any group or individual who can affect or is affected by the achievement of

² Stakeholder is defined as "any group or individual who can affect or is affected by the achievement of the organization's objectives" (Freeman, 1984; Jones, 1995; Kreiner & Bhambri, 1988) (reported in Mitchell, Agle, & Wood, 1997, p. 869). According to European Foundation for Quality Management (EFQM) (2010, p. 2), stakeholder includes the "person, group or organisation that has a direct or indirect stake or interest in the organisation because it can either affect the organisation or be affected by it."

(Bergquist, Foley, Garvare, & Johansson, 2000, p. 517; Edgeman, Dahlgaard-Park, Dahlgaard, & Scherer, 1999, p. 49)

• "Business aim (that circumstance where all stakeholders are satisfied), and [total] quality management ... as the means of achieving that aim" (Foley, 2004, p. 13).

BE is also known as Organisational Excellence. The term "Organisational Excellence" emerged to imply inclusion of public and not-for-profit organisations (Dalrymple et al., 1999; McAdam, 2000). Many BE concepts, frameworks and models were developed since 1980s, mainly as a result of quality movement in Japan and USA (Adebanjo & Mann, 2008b). Examples of the BE concepts and models include Peter's and Waterman's eight excellence attributes (1982), Peter's and Austin's excellence model (1985), Baldrige Criteria for Performance Excellence (CPE) (first developed in 1987), European Foundation for Quality Management (EFQM) Excellence Model (first developed in 1992), Kanji's BE Model (2001), and Dahlgaard's and Dahlgaard-Park's 4P excellence model (2004).

This research focuses on BEMs used within guality and BE award programmes, due to the validity and wide usage of these models. This type of BEM includes the Baldrige CPE (NIST, 2010) and the EFQM Excellence Model (EFQM, 2009). According to Grigg and Mann (2008b, p. 1176), various researchers have "statistically tested and validated" the design, criteria and items of the major BEMs using techniques including simple bivariate correlation (Saunders & Mann, 2005); path analysis (Flynn & Saladin, 2001); and Covariance Based Structural Equation Modelling (Lee, Rho, & Lee, 2003; Wilson & Collier, 2000). As reported by Grigg and Mann (2008a, p. 234; 2008b, p. 1173), BEMs are adopted for Quality / BE Awards by more than 80 countries. Further research by the author of this thesis in 2010 (which was commissioned by Baldrige Performance Excellence Program, NIST, USA) revealed that there were 94 national Quality / BE Awards used in 83 countries worldwide. Appendix 1 lists the awards together with their adopted BEMs and administrative organisations. This information is important for the users and potential users of BEMs as well as the organisations entrusted to administer the national quality / BE award. This list was developed based on data from Quality / BE award administrators, an internet search conducted between 9th and 13th of January 2010 and latest updated on November 2010, as well as data from a comprehensive literature review (Calingo, 2002; Mavroidis, Toliopoulou, & Agoritsas, 2007; Sharma

& Kodali, 2008; Tan, 2002; Tan, Wong, Mehta, & Khoo, 2003). An early version of this list is available on the Baldrige Performance Excellence Program website (refer to Mohammad & Mann, 2010).

The data presented in Appendix 1 indicates that the EFQM Excellence Model and the Baldrige CPE are the two most widely used BEMs. The EFQM Excellence Model is used in 30 countries on two continents - Europe (e.g. Austria, Northern Ireland, Sweden, Italy, and Portugal) and Asia (India, Turkey and United Arab Emirates). Meanwhile, the Baldrige CPE is used in 8 countries on four continents, including Northern America (United States of America), Asia (Hong Kong, Indonesia, Philippines, Thailand and Sri Lanka), Oceania (New Zealand), and Europe (Sweden). Many countries are also using their own national bespoke model, such as Japan (Deming Prize) and Australia (Australian BE Award). In addition, some countries have more than one national quality / BE awards, such as India, Japan, Malaysia, United Arab Emirates, Greece and Hungary. In Sweden, organisations can choose one of the following BEMs when applying for Swedish Quality Award: (1) the Baldrige CPE (see item 6 in Appendix 1); (2) the EFQM Excellence Model (see item 34 in Appendix 1); or the Swedish bespoke model (see item 80 in Appendix 1). In contrast, Bahrain does not have any national quality / BE award to date, but many public sector organisations are using a bespoke model based on the EFQM Excellence Model, which is administered by the Bahrain Centre for Excellence (see item 60 in Appendix 1).

In general, a BEM can be considered as a TQM framework (Adebanjo, 2001; Adebanjo & Mann, 2008b; Bou-Llusar, Escrig-Tena, Roca-Puig, & Beltran-Martin, 2009; Dale, Van der Wiele, & Van Iwaarden, 2007a). More specifically, the BEM can be described as a non-prescriptive organisational framework based on several main criteria that can be categorised as 'enabler' and 'result' (Dahlgaard-Park, 2008; EFQM, 2003; NIST, 2010). Dahlgaard-Park (2008) summarised that:

The "Enabler" criteria cover what an organisation does. The "Result" criteria cover what an organisation achieves. "Enablers" cause "Results". (p. 106)

Table 2.2 shows the common criteria of BEMs based on comparison of major BEMs and previous research. The descriptions of the common enabler criteria of BEM and their core values and concepts are presented in Table 2.3. These core values and concepts of BEM are based on attributes, beliefs and/or behaviours of high performing organisations (NIST, 2010; New Zealand Business Excellence Foundation [NZBEF], 2009).

Category	Common Criteria	Baldrige CPE 2011 - 2012 (NIST, 2010)	EFQM Excellence Model 2010 (EFQM, 2009)	Bohoris (1995)	Puay, Tan, Xie, & Goh (1998)	Tan (2002)
Enablers	1. Leadership and social responsibilities	1.Leadership	1.Leadership	1.Leadership	1.Leadership 2.Impact on	1.Leadership system 2.Impact on society
	2. Strategy	2.Strategic planning	2.Strategy	2.Strategy and policy	3.Strategy and policy	3.Strategy and policy planning
	3. Customer focus	3.Customer focus	3.Processes, products and	3.Customer management and satisfaction	4.Customer management and satisfaction	4.Customer management and satisfaction
	4. Process management	4.Operations focus	services	4.Process quality	5.Process quality	5.Process management
	5. Workforce focus	5.Workforce focus	4.People	5.Human resource management	6.Human resource management	6.People management
	6. Partnership and resources	6.Measurement, analysis and knowledge	5. Partnership and resources	6.Resources Management	7.Resources management	7.Resources 8.Information and analysis
		management			8.Suppliers / partners management and performance	9.Performance and management of suppliers / partners
Results	7. Results	7. Results	6.People results 7.Customer results 8.Society results 9 Key results	7.Results	9.Results	10.Business results

Table 2.2: Common criteria of BEMs based on comparison of major BEMs and previous research

Table 2.3: Descriptions of the common enabler cri	iteria of BEM and their core values
---	-------------------------------------

and concepts

Common enabler criteria of BEM	Brief descriptions	Core values and	
 Leadership and social responsibilities 	Focuses on leader's behaviour and leadership system in shaping the future and building culture of excellence in the organisation (EFQM, 2009; NIST, 2010; Puay, et al., 1998; Standards, Productivity, and Innovation Board Singapore [SPRING], 2007), as well as, organisation's governance systems, responsibilities and contribution to society, community and environment (EFQM, 2009; NIST, 2010; Puay, et al., 1998; Tan, 2002).	Leading with vision, inspiration and integrity; systems perspective; and taking responsibility for a sustainable future (EFQM, 2009; NIST, 2010).	
2. Strategy	Focuses on how the organisation develops, communicates, deploys, measures, monitors, reviews and / or improves its strategy to achieve organisational success and sustainability (EFQM, 2009; NIST, 2010; Puay, et al., 1998; Tan, 2002).	Visionary leadership; focus on the future; nurturing creativity and innovation; and agility (EFQM, 2009; NIST, 2010).	
3. Customer focus	Focuses on how the organisation determines customers and market needs and expectations; builds relationships with customers; uses customer information to improve and identify opportunities for innovation; and determines customer satisfaction for long-term marketplace success (EFQM, 2009; NIST, 2010; Puay et al., 1998; SPRING, 2007; Tan, 2002).	Customer-driven excellence; focus on results; and creating value (EFQM, 2009; NIST, 2010).	
4. Process management	Focuses on the design, management, evaluation, and improvement of various work systems and work processes in the organisation in order to fully satisfy, and generate increasing value for customers and other stakeholders (EFQM, 2009; NIST, 2010; Puay et al., 1998; Tan, 2002).	Organisational learning; nurturing creativity and innovation; focus on results and creating value; agility; and systems perspective (EFQM, 2009; NIST, 2010).	
5. Workforce focus	Focuses on how the organisation engages, manages, values, recognises and develops the workforce to utilise its maximum potential in alignment with the organisation's overall mission, strategy and plan (EFQM, 2009; NIST, 2010; Tan, 2002).	Personal learning; valuing workforce members; focus on results and creating value; succeeding through people; and nurturing creativity and innovation (EFQM, 2009; NIST, 2010).	
6. Partnership and resources	Focuses on how the organisations plan, manage, measure, analyse and improve external partnerships, suppliers and internal resources (such as information and knowledge, financial, materials, natural resources, buildings, equipment, technology, and intellectual property) in order to support strategy and the effective operation of work processes (EFQM, 2009; Puay et al., 1998; Tan, 2002).	Building partnerships; management by fact; focus on results and creating value; and nurturing creativity and innovation (EFQM, 2009; NIST, 2010)	

Note: These descriptions were included in the proposed guidance model (see Supplement B in Appendix 8, page A8.11)

Dahlgaard-Park (2008) has analysed the EFQM Excellence Model from six management control perspectives involving: (1) Bureaucratic and mechanic view of control, (2) Cybernetic view of control, (3) Agency view of control, (4) Human resource view of control, (5) Contingency view of control and (6) Cultural view of control. In addition, Dahlgaard-Park (2008, p. 111) found that "all the criteria [of the model] show more or less interrelationships with the six management control theories" and concluded that "the model can be considered as a holistic and integrative approach, where strategic, managerial and operational control processes are integrated in the model". On the other hand, Dahlgaard-Park (2008) also highlighted the disadvantages and limitations of the EFQM Excellence Model, which includes: (1) less attention is given to contextual / contingency factors; (2) not being able to include all possible variables and all aspects of real situation due to the nature of a model that only represent a generalised and simplified version of a reality; and (3) the actual practice is not always consistent with the expectation when using the model.

BEMs are widely used as a basis for evaluating the performance of Quality / BE Award applicants as well as to identify the winners of the award (Adebanjo & Mann, 2008b; Tan et al., 2003). Although numerous organisations have participated in the BE award programmes, the main purpose of using the BEMs should be for organisational performance improvement rather than merely receiving the award (Dahlgaard-Park & Dahlgaard, 2007).

Most of the previous studies into BEMs mainly focused on the design and/or validation of BEMs (Dahlgaard-Park & Dahlgaard, 2007; Evans & Jack, 2003; Flynn & Saladin, 2001; Husain, Abdullah, Idris, & Sagir, 2001; Jayamaha, Grigg, & Mann, 2009), the usage of BEMs for quality / BE awards (Eriksson & Garvare, 2005; Grigg & Mann, 2008b), the usage of BEMs for organisational assessment (Ritchie & Dale, 2000; Shergold & Reed, 1996; Williams, Bertsch, van der Wiele, Van Iwaarden, & Dale, 2006) as well as the usage of BEMs for benchmarking and best practices (Adebanjo & Mann, 2008a; Mann & Grigg, 2004). Minimal literature used BEMs as a guiding and/or overarching frameworks for selecting and managing multiple improvement initiatives (such as, Brown & Pemberton Planning Group Ltd., 2008; Saunders & Mann, 2007).

In relation to the usage of BEMs as an overarching framework, Jamie Ambrosi, Deputy Director of Baldrige Performance Excellence Program (as cited in Mann, 2011) highlighted that:

I think where organizations get off track is when they think Baldrige is just an initiative, rather than a model for organizing and managing the enterprise and all its initiatives. If Baldrige is reduced to an initiative, rather than an overall model and a way of thinking, then organizations can say they have done it and moved on. We see this all the time. But in organizations that embrace the Baldrige Framework as an overarching model, they never move beyond it. This includes very high-performing organizations, including our Award recipients. (p.109)

Supporting the same issue, Joe Goasdoue, Chief Executive of the British Quality Foundation (as cited in Francis, 2010), explains that:

While there are numerous management tools and techniques commonly used, the EFQM excellence model provides a holistic view of the organisation and it can be used to determine how these different methods fit together and complement each other. The model can therefore be used in conjunction with any number of these tools, based on the needs and function of the organisation, as an overarching framework for developing sustainable excellence. (p.30)

The proposed guidance model utilised the BEM as a guiding framework for selecting and managing multiple improvement initiatives. Part of the model shows some examples of the most common improvement initiatives that can be adopted towards BE, which are arranged according to the common enabling criteria of BEMs and levels of BE maturity. In this study, common enabling criteria of BEMs represent areas for improvement. This feature can help organisations to choose appropriate improvement initiatives by narrowing down the options according to the areas for improvement and BE maturity.

References

- Abarca, D. (1998). Implementing ISO 9000 and ISO 14000 concurrently. *Pollution Engineering*, *30*(10), 46-48.
- Ackoff, R. L. (1999). *Re-creating the Corporation: A design of organisations for the 21st century*. Oxford: Oxford University Press.
- Adebanjo, D. (2001). TQM and business excellence: is there really a conflict? *Measuring Business Excellence, 5*(3), 37-40.
- Adebanjo, D., & Mann, R. (2008a). Sustainability of benchmarking networks: A case-based analysis. *Total Quality Management & Business Excellence, 19*(1–2), 107–122.
- Adebanjo, D., & Mann, R. (2008b). Business Excellence. *BPIR Management Brief, 4*(6), 1-16.
- Adebanjo, D., & Mann, R. (2008c). Benchmarking. *BPIR Management Brief, 4*(5), 1-12.
- Adomaityte, J. (2011). *Adaptation of service excellence awards in Europe.* Unpublished master thesis, Erasmus University.
- Allen, I. E., & Seaman, C. A. (2007). Likert scales and data analyses. *Quality Progress, 40*(7), 64-65.
- Antony, J. (2007). Six Sigma: a strategy for supporting innovation in pursuit of business excellence. International Journal of Technology Management, 37(1-2), 8-12.
- Antony, J., & Banuelas, R. (2002). Key ingredients for the effective implementation of Six Sigma program. *Measuring Business Excellence, 6*(4), 20-27.
- Arnheiter, D. A., & Maleyeff, J. (2005). The integration of lean management and Six Sigma. *The TQM Magazine, 17*(1), 5-18.
- Ashmos, D. P., & Huber, G. P. (1987). The systems paradigm in organization theory: Correcting the record and suggesting the future. *The Academy of Management Review, 12*(4), 607-621.
- Basu, R. (2004a). Six-Sigma to operational excellence: Role of tools and techniques. *International Journal of Six Sigma and Competitive Advantage, 1*(1), 44-64.
- Basu, R. (2004b). *Implementing quality: A practical guide to tools and techniques*. London: Thomson.
- Basu, R., & Wright, J. N. (2005). *Total operations solutions*. Oxford: Butterworth-Heinemann.

- Baxter, L. F., & MacLeod, A. M. (2008). *Managing performance improvement*. New York: Routledge.
- Bazerman, M. H., & Moore, D. A. (2009). Judgment in managerial decision making (7th ed.). New Jersey: John Wiley & Sons Inc.
- Bendell, T. (2000). The implications of the changes to ISO 9000 for organisational excellence. *Measuring Business Excellence, 4*(3), 11-14.
- Bendell, T. (2005). Structuring business process improvement methodologies. *Total Quality Management & Business Excellence, 16*(8-9), 969-978.
- Benson, P. G., Saraph, J. V., & Schroeder, R. G. (1991). The effects of organisational context on quality management: An empirical investigation. *Management Science*, 37(9), 1107-1124.
- Bergquist, B., Foley, K., Garvare, R., & Johansson, P. (2000). Reframing Quality Management. In K. J. Foley & P. Hermel (Eds.), *The Theories and Practices* of Organizational Excellence: New Perspectives (pp. 501-538). Sydney: SAI Global.
- Bohoris, G. A. (1995). A comparative assessment of some major quality awards. International Journal of Quality & Reliability Management, 12(9), 30-43.
- Bou-Llusar, J. C., Escrig-Tena, A. B., Roca-Puig, V., & Beltran-Martin, I. (2009). An empirical assessment of the EFQM Excellence Model: Evaluation as a TQM framework relative to the MBNQA Model. *Journal of Operations Management, 27*, 1-22.
- Bourque, L. B., & Fielder, E. P. (2003). *How to conduct self-administered and mail surveys* (2nd ed.). California: Sage Publications.
- Brown & Pemberton Planning Group Ltd. (2008). Report on Business Excellence Schemes. Retrieved 23 April, 2009, from http://www.qualityplanning.org.nz /qp-research/business-excellence-scheme/ index.php
- Bryman, A., Becker, S., & Sempik, J. (2008). Quality criteria for quantitative, qualitative and mixed methods research: A view from social policy. *International Journal of Social Research Methodology*, *11*(4), 261-276.

Burnell, S. (2008). Future perfect? Quality World, January, 22.

- Business Performance Improvement Resource (BPIR). (2008a). Benchmarking and performance excellence self-assessment (Baldrige criteria). Retrieved 28 July, 2008, from http://members.bpir.com/site/sassessments/BPESBC.pdf
- Business Performance Improvement Resource (BPIR). (2008b). Benchmarking and performance excellence self-assessment (EFQM excellence model). Retrieved 28 July, 2008, from http://members.bpir.com/site/sassessments/ BPESEFQM.pdf

- Business Performance Improvement Resource (BPIR). (2009). Strategies, tools and techniques. Retrieved 12 February, 2009, from http://www.bpir.com/ component/option,com_bpirmembers/Itemid,99/success,strategies/
- Byrne, G., Lubowe, D., & Blitz, A. (2007). Using a Lean Six Sigma approach to drive innovation. *Strategy and Leadership, 35*(2), 5-10.
- Cagliano, R., & Spina, G. (2000). How improvement programmes of manufacturing are selected. *International Journal of Operations & Production Management, 20*(7), 772-791.
- Calingo, L. M. R. (2002). National quality and business excellence awards: Mapping the field and prospects for Asia. In L. M. R. Calingo (Ed.), *The quest for* global competitiveness through national quality and business excellence awards (pp. 3-18). Tokyo: Asian Productivity Organisation.
- Cameron, K. S., & Barnett, C. K. (1999). Organizational quality as a cultural variable: An empirical investigation of quality culture, processes and outcomes. In R. E. Cole & W. R. Scott (Eds.), *The Quality Movement and Organization Theory* (pp. 271-294). California: Sage Publications Inc.
- Canez, L. E., Platts, K. W., & Probert, D. R. (2000). Developing a framework for make-or-buy decisions. *International Journal of Operations & Production Management, 20*(11), 1313-1330.
- Capon, C. (2004). Understanding organisational context: Inside and outside organisations (2nd ed.). Harlow: Prentice Hall Financial Times.
- Caracelli, V. J., & Riggin, L. J. C. (1994). Mixed-method evaluation: Developing quality criteria through concept mapping: Mixed-Method Collaboration. *Evaluation Practice, 15*(2), 139-152.
- Casadesus, M., & Karapetrovic, S. (2005). An empirical study of the benefits and costs of ISO 9001: 2000 compared to ISO 9001/2/3: 1994. *Total Quality Management & Business Excellence, 16*(1), 105-120.
- Clark, T. A. R. (2004). The fashion of management fashion: A surge too far? *Organization, 11*(2), 297-306.
- Cobb, C. G. (2003). From quality to business excellence: A systems approach to management. Wisconsin: ASQ Quality Press.
- Consortium for Excellence in Higher Education (CEHE). (2003). Linking the EFQM Excellence Model to other management models and tools. Retrieved 17 July 2008, from http://www.shu.ac.uk/research/integralexcellence/downloads/ managementmodels.pdf
- Crawford, N. (2007). Performance Management. BPIR Management Brief, 4(2), 1-10.

- Creswell, J. W. (2009). *Research design: Qualitative, quantitative, and mixed methods approaches* (3rd ed.). Thousand Oaks, CA: Sage Publications.
- Creswell, J. W., & Clark, V. L. P. (2007). *Designing and conducting mixed methods research*. Thousand Oaks, CA: Sage Publications.
- Creswell, J. W., & Clark, V. L. P. (2011). *Designing and conducting mixed methods research* (2nd ed.). Thousand Oaks, CA: Sage Publications.
- Cullen, J., O'Connor, M., & Mangan, J. (2004). Matching management tools and techniques with management challenges. *Strategy & Leadership*, 32(3), 27-30.
- Dahlgaard-Park, S. M. (2008). Reviewing the European excellence model from a management control view. *The TQM Journal, 20*(2), 98-119.
- Dahlgaard-Park, S. M., & Dahlgaard, J. J. (2007). Excellence 25 years evolution. *Journal of Management History, 13*(4), 371-393.
- Dahlgaard, J. J., & Dahlgaard-Park, S. M. (2004). The 4P quality strategy for breakthrough and sustainable development. *European Quality*, *10*(4), 6-20.
- Dale, B. G. (2007). Tools and techniques: An overview. In B. G. Dale, T. Van der Wiele & J. Van Iwaarden (Eds.), *Managing quality* (5th ed., pp. 336-381).Malden, MA: Blackwell Publishing.
- Dale, B. G., & Lascelles, D. M. (1997). Total quality management adoption: Revisiting the levels. *The TQM Magazine, 9*(6), 418-428.
- Dale, B. G., & Lascelles, D. M. (2007). Levels of TQM adoption. In B. G. Dale, T.
 Van der Weile & J. Van Iwaarden (Eds.), *Managing Quality* (5th ed., pp. 111-126). Malden, MA: Blackwell Publishing Ltd.
- Dale, B. G., & McQuater, R. (1998). *Managing business improvement and quality: Implementing key tools and techniques*. Oxford: Blackwell Publishers.
- Dale, B. G., Van der Wiele, A., & Van Iwaarden, J. D. (2007a). Self-assessment, models and quality awards. In B. G. Dale, T. Van der Wiele & J. Van Iwaarden (Eds.), *Managing Quality* (5th ed., pp. 531-558). Malden, MA: Blackwell Publishing.
- Dale, B. G., Van der Wiele, A., & Van Iwaarden, J. D. (2007b). TQM: An overview.In B. G. Dale, T. Van der Wiele & J. Van Iwaarden (Eds.), *Managing Quality* (pp. 3-38). Malden, MA: Blackwell Publishing.
- Dale, B. G., Van der Wiele, A., & Van Iwaarden, J. D. (2007c). Teams and teamwork. In B. G. Dale, A. van der Wiele & J. D. van Iwaarden (Eds.), *Managing Quality* (pp. 510-530). Malden, MA: Blackwell Publishing.

- Dalrymple, J., Edgeman, R. L., Finster, M., Guerrero-Cusumano, J. L., Hensler, D. A., & Parr, W. C. (1999). Next-generation quality management: multinational, multidisciplinary and performance-focused. *The TQM Magazine, 11*(3), 138-141.
- Davenport, T. H., Prusak, L., & Wilson, H. J. (2003). *What's the big idea? Creating and capitalizing on the best management thinking*. Boston: Harvard Business School Press.
- Dean, J. W., & Bowen, D. E. (1994). Management theory and total quality: improving research and practice through theory development. *The Academy of Management Review, 19*(3), 392-418.
- Dellinger, A. B., & Leech, N. L. (2007). Toward a unified validation framework in mixed methods research. *Journal of Mixed Methods Research*, 1(4), 309-332.
- Doctoral Research Committee (DRC). (2011). *Handbook for doctoral study Version* 7. Palmerston North: Graduate Research School, Massey University.
- Donaldson, L. (2001). *The contingency theory of organizations*. California, CA: Sage Publications.
- Douglas, A., & Glen, D. (2000). Integrated management systems in small and medium enterprises. *Total Quality Management*, *11*(4/5), S686-S690.
- Edgeman, R. L., Dahlgaard-Park, S. M., Dahlgaard, J. J., & Scherer, F. (1999). On leaders and leadership. *Quality Progress, 32*(10), 49-54.
- Eisenhardt, K. M. (1999). Strategy as strategic decision making. *Sloan Management Review, 40*(3), 65-72.
- Eisenhardt, K. M., & Zbaracki, M. J. (1992). Strategic decision making. *Strategic Management Journal, 13*, 17-37.
- El-Kafafi, S. (2006a). Building bridges between organisational change and quality management in New Zealand. *International Journal of Management Science and Engineering Management*, 1(1), 53-62.
- El-Kafafi, S. (2006b). TQM models and their effectiveness in New Zealand water utilities services. *The TQM Magazine, 18*(5), 440-454.
- English, G. (1998). *Phoenix without the ashes: Achieving organizational excellence through common sense management* Boca Raton: St. Lucie Press.
- Eriksson, H., & Garvare, R. (2005). Organisational performance improvement through quality award process participation. *International Journal of Quality & Reliability Management*, 22(9), 894-912.
- European Foundation for Quality Management (EFQM). (2003). Introducing excellence. Brussels: EFQM.

- European Foundation for Quality Management (EFQM). (2009). Introducing the EFQM Excellence Model 2010. Retrieved 30 November, 2010, from http://www.efqm.org/en/PdfResources/EFQMModel_Presentation.pdf
- European Foundation for Quality Management (EFQM). (2010). *EFQM user guide: Understanding an organisation's stakeholders*. Brussels: EFQM.
- Evans, J. R. (2008). *Quality and performance excellence: Management, organization, and strategy* (5th ed.). Mason, OH: Thomson South-Western.
- Evans, J. R., & Jack, E. P. (2003). Validating key results linkages in the Baldrige performance excellence model. *Quality Management Journal, 10*(2), 7-24.
- Flynn, B. B., & Saladin, B. (2001). Further evidence on the validity of the theoretical models underlying the Baldrige criteria. *Journal of Operations Management*, 19(6), 617-652.
- Foley, K. J. (2004). *Five essays on Quality Management*. Sydney: Consensus Books.
- Foley, K. J. (2010). Third generation quality management: From atoms to bits, or quality management in the knowledge society. Paper presented at the QUALCON 2010 International Conference in Quality Management, Canberra.
- Francis, F. (2010). One hundred to one. Quality World, 36(1), 26-31.
- Gharajedaghi, J. (2006). Systems thinking Managing chaos and complexity: A platform for designing business architecture (2nd ed.). Burlington, MA: Butterworth-Heinemann
- Goetsch, D. L., & Davis, S. B. (2006). *Quality Management* (5th ed.). New Jersey: Prentice Hall.
- Greatbatch, D., & Clark, T. (2005). *Management speak: Why we listen to what management gurus tell us*. London: Routledge
- Grigg, N., & Mann, R. (2008a). Promoting excellence An international study into creating awareness of business excellence models. *The TQM Journal, 20*(3), 233-248.
- Grigg, N., & Mann, R. (2008b). Review of the Australian Business Excellence Framework: A comparison of national strategies for designing, administering and promoting Business Excellence Frameworks. *Total Quality Management and Business Excellence, 19*(11), 1173-1188.
- Grigg, N. P. (2004). *Developing statistical thinking in the food industry.* Unpublished doctoral thesis, University of Strathclyde, United Kingdom.
- Grigg, N. P., & Walls, L. (1999). The use of statistical process control in food packing. *British Food Journal, 101*(10), 763-784.

- Guba, E. G. (1990). The alternative paradigm dialog. In E. G. Guba (Ed.), *The paradigm dialog* (pp. 17-27). Newbury Park, CA: Sage.
- Harrington, H. J. (1995). Continuous versus breakthrough improvement: Finding the right answer. *Business Process Re-engineering and Management Journal, 1*(3), 31-49.
- Harrington, H. J., & Harrington, J. S. (1995). *Total Improvement Management: The next generation in performance improvement*. New York, NY: McGraw-Hill Inc.
- Harrington, H. J., & Lomax, K. C. (2000). *Performance improvement methods: Fighting the war on waste.* New York, NY: McGraw-Hill.
- Hayes, R., Pisano, G., Upton, D., & Wheelwright, S. (2005). *Operations, strategy, and technology: Pursuing the competitive edge*. Hoboken, NJ: Wiley.
- Hellsten, U., & Klefsjö, B. (2000). TQM as a management system consisting of values, techniques and tools. *The TQM Magazine, 12*(4), 238-244.
- Hendra, I. (2010). Horses for courses picking your winner when it comes to quality improvement systems. In L. Nikoloff (Ed.), *QNewZ* (January-February, pp. 11-14). Palmerston North: New Zealand Organisation for Quality
- Hicks, M. J. (2004). *Problem solving and decision making: Hard, soft and creative approaches* (2nd ed.). London: Thomson Learning.
- Hofstede, G. (1980). Culture's consequences: International differences in workrelated values. Newbury Park: Sage.
- Hofstede, G., & Bond, M. H. (1988). The Confucius connection: From cultural roots to economic growth. *Organizational Dynamics, 16*(4), 5-21.
- Huber, G. P. (1980). *Managerial decision making*. Illinois: Scott, Foresman and Company.
- Husain, N., Abdullah, M., Idris, F., & Sagir, R. M. (2001). The Malaysian Total Performance Excellence Model: A conceptual framework. *Total Quality Management & Business Excellence*, *12*(7), 926-931.
- Imai, M. (1986). *Kaizen: The key to Japan's competitive success*. New York, NY: McGraw-Hill.
- Imai, M. (1997). Gemba kaizen: A commonsense, low-cost approach to management. Boston, MA: McGraw-Hill.
- Innes, J. (2005). A conceptual framework for starting the Business Excellence (BE) journey in New Zealand service organisations. Unpublished doctoral thesis, Massey University, Palmerston North.

- Institute of Risk Management (IRM), Association of Insurance and Risk Managers (AIRMIC), & National Forum for Risk Management in the Public Sector (ALARM). (2002). A risk management standard. Retrieved 8 December, 2008, from http://www.theirm.org/publications/documents/Risk_Management _Standard_ 030820.pdf
- International Organization for Standardization (ISO). (2000). ISO 9000 : Quality management systems Fundamentals and vocabulary (2nd ed.). Geneva: ISO.
- Jackson, M. C. (2000). Systems approaches to management. New York: Kluwer Academic / Plenum Publishers.
- Jackson, M. C. (2003). Systems thinking: Creative holism for managers. Chichester: John Wiley & Sons Ltd.
- Jamieson, S. (2004). Likert scales: how to (ab)use them. *Medical Education, 38*(12), 1217-1218.
- Jayamaha, N., Grigg, N., & Mann, R. (2009). A study of the validity of three major business excellence models in the Asia Pacific region. *Total Quality Management and Business Excellence, 20*(11), 1213-1227.
- Johnson, R. A., Kast, F. E., & Rosenzweig, J. E. (1964). Systems theory and management. *Management Science, 10*(2), 367-384.
- Kanji, G. K. (2001). Forces of excellence in Kanji's Business Excellence Model. *Total Quality Management & Business Excellence, 12*(2), 259-272.
- Karapetrovic, S., & Willborn, W. (1998). Integration of quality and environmental management systems. *The TQM Magazine, 10*(3), 204-213.
- Kast, F. E., & Rosenzweig, J. E. (1972). General systems theory: Applications for organization and management. *The Academy of Management Journal*, 15(4), 447-465.
- Kast, F. E., & Rosenzweig, J. E. (1985). *Organization and management: A systems and contingency approach* (4th ed.). New York, NY: McGraw-Hill.
- Koontz, H., & O'Donnell, C. (1976). *Management: A systems and contingency analysis of managerial functions* (6th ed.). Tokyo: McGraw-Hill.
- Kwok, K. Y., & Tummala, V. M. R. (1998). A quality control and improvement system based on the total control methodology (TCM). *International Journal of Quality & Reliability Management, 15*(1), 13-48.
- Lawrence, P. R., & Lorsch, J. W. (1967). *Organization and environment: Managing differentiation and integration*. Massachusetts: Harvard University Press.

- Lee, S. M., Rho, B. H., & Lee, S. G. (2003). Impact of Malcolm Baldrige National Quality Award Criteria on organizational quality performance. *International Journal of Production Research*, 41(9), 2003-2020.
- Lewins, A., & Silver, C. (2007). Using software in qualitative research: A step-bystep guide. London: SAGE Publications.
- Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic inquiry*. Newbury Park, CA: Sage Publications.
- Litwin, M. S. (2003). *How to assess and interpret survey psychometrics* (2nd ed.). California: Sage Publications.
- Mann, R. (2008a). Global survey on business improvement and benchmarking. Retrieved 25 April, 2008, from http://www.bpir.com/gbn-survey-bpir.com
- Mann, R. (2008b). Revisiting a TQM research project: The quality improvement activities of TQM. *Total Quality Management & Business Excellence, 19*(7), 751-761.
- Mann, R., Adebanjo, D., Laosirihongthong, T., & Punnakitikashem, P. (2011). Awareness and impact of business excellence in Asia. *Total Quality Management & Business Excellence, 22*(11), 1237-1258.
- Mann, R., & Grigg, N. (2004). Helping the Kiwi to fly: Creating world-class organisations in New Zealand through a benchmarking initiative. *Total Quality Management & Business Excellence, 15*(5-6), 707-718.
- Mann, R. S. (1992). *The development of a framework to assist in the implementation of Total Quality Management (TQM).* Unpublished doctoral thesis, University of Liverpool.
- Mann, R. S. (2011). *Impact of business excellence / quality awards on enterprises.* Tokyo: Asian Productivity Organization.
- Mann, R. S., Abbas, A., & Kohl, H. (2010). *Global survey on business improvement and benchmarking*. Berlin: Global Benchmarking Network.
- Mann, R. S., & Kehoe, D. F. (1994). The quality improvement activities of Total Quality Management (Paper 1). *Quality World Technical Supplement* (March), 43-56.
- Mavroidis, V., Toliopoulou, S., & Agoritsas, C. (2007). A comparative analysis and review of national quality awards in Europe: Development of critical success factors. *The TQM Journal*, *19*(5), 454-467.
- Maxwell, J. A. (2002). Understanding and validity in qualitative research. In A. M.Huberman & M. B. Miles (Eds.), *The qualitative researcher's companion* (pp. 2). Thousand Oaks, CA: Sage Publications.

- McAdam, R. (2000). Three leafed clover?: TQM, organisational excellence and business improvement. *The TQM Journal, 12*(5), 314-320.
- McAdam, R., & Kelly, M. (2002). A business excellence approach to generic benchmarking in SMEs. *Benchmarking: An International Journal, 9*(1), 7-27.
- Miller, D., & Hartwick, J. (2002). Spotting management fads. *Harvard Business Review, 80*(10), 26-27.
- Mitchell, R. K., Agle, B. R., & Wood, D. J. (1997). Toward a theory of stakeholder identification and salience: defining the principle of who and what really counts. *Academy of Management Review, 22*(4), 853-886.
- Mohammad, M., & Mann, R. (2010). National Quality / Business Excellence Awards in different countries. Retrieved 8 November, 2010, from http://www.nist.gov/baldrige/community/upload/National_Quality_Business_E xcellence_Awards_in_Different_Countries.xls
- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2008). Multiple Initiatives Implementation in the Journey towards Organisational Excellence: Issues and Future Research. In L. Coburn, J. Kenny & P. Castka (Eds.), *Proceedings of the New Zealand Organisation for Quality (NZOQ) Conference 2008* (pp. 67-69): New Zealand Organisation for Quality
- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2009a). Selection of business improvement initiatives towards achieving business excellence: An initial conceptual model. *Proceedings of the International Conference on Quality, Productivity and Performance Measurement*. Putrajaya: Universiti Kebangsaan Malaysia.
- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2009b). Selection of quality improvement initiatives: An initial conceptual model. *Journal of Quality Measurement and Analysis, 5*(2), 1-14.
- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2010a). The right improvement initiative for the right situation: A contextual and systems approach. In H. Ishii, T. Nose, M. Gen & S. Shiode (Eds.), *Proceedings of the 40th International Conference on Computers and Industrial Engineering* (pp. 1-6). Hyogo: Institute of Electrical and Electronics Engineers (IEEE).
- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2010b). Selecting the right organisational improvement initiative. In L. Nikoloff (Ed.), *QNewZ Magazine of the New Zealand Organisation for Quality* (November-December, pp. 9-11). Palmerston North: New Zealand Organisation for Quality.

- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2011a). Business Excellence Model: An overarching framework for managing and aligning multiple organisational improvement initiatives. *Total Quality Management and Business Excellence*, 22(11), 1213-1236.
- Mohammad, M., Mann, R., Grigg, N., & Wagner, J. P. (2011b). Selecting appropriate organisational improvement initiatives: A five steps approach. In S. K. M. Ho (Ed.), *Proceedings of the 15th International Conference on ISO and TQM* (pp. 1-6). Hong Kong: Asia Pacific Business Excellence Standard Academy.
- Mohammad, M., Osman, M. R., Yusuff, R. M., & Ismail, N. (2005). Strategies and critical success factors for integrated management systems implementation.
 In M. B. Durmusoglu & C. Kahraman (Eds.), *Proceedings of the 35th International Conference on Computers and Industrial Engineering* (Vol. 2, pp. 1391-1396). Istanbul: Istanbul Technical University.
- Morse, J. M. (2003). Principles of mixed methods and multimethod research design.
 In A. Tashakkori & C. Teddlie (Eds.), *Handbook of Mixed Methods in Social* and Behavioral Research (pp. 189-208). California: Sage Publications.
- National Institute of Standards and Technology (NIST). (2008). 2009-2010 Criteria for performance excellence. Retrieved 8 December, 2008, from http://www.baldrige.nist.gov/PDF_files/2009_2010_Business_Nonprofit_Crite ria.pdf
- National Institute of Standards and Technology (NIST). (2010). 2011-2012 Criteria for performance excellence. Retrieved 23 December, 2010, from http://www.nist.gov/baldrige/publications/upload/2011_2012_Business_Nonp rofit_Criteria.pdf
- Nave, D. (2002). How to compare Six Sigma, Lean and the Theory of Constraints: A framework for choosing what's best for your organization. *Quality Progress*, *35*(3), 73-78.
- Neuman, W. L. (2006). Social research methods: Qualitative and quantitative approaches (6th ed.). Boston: Pearson.
- New Zealand Aluminium Smelters (NZAS). (2007). New Zealand Business Excellence Award application report. Invercargill: NZAS.
- New Zealand Business Excellence Foundation (NZBEF). (2009). *What is business excellence Part one*. North Shore: New Zealand Business Excellence Foundation.

- Newman, I., Ridenour, C. S., Newman, C., & DeMarco-Jr., G. M. P. (2003). A typology of research purposes and its relationship to mixed methods. In A. Tashakkori & C. Teddlie (Eds.), *Handbook of Mixed Methods in Social and Behavioral Research* (pp. 167-188). California: Sage Publications.
- O'Cathain, A. (2010). Assessing the quality of mixed methods research. In A. Tashakkori & C. Teddlie (Eds.), SAGE Handbook of Mixed Methods in Social & Behavioral Research (2nd ed., pp. 531-555). Thousand Oaks, CA: SAGE Publications.
- Paropkari, A. (2011). Estimating ROI for a process improvement program. Retrieved 12 December, 2011, from http://www.isixsigma.com/index.php? option=com_k2&view=item&id=797:&Itemid=187
- Patton, M. Q. (2002). *Qualitative research and evaluation methods* (3rd ed.). California: Sage Publications.
- Peters, T. J., & Austin, N. (1985). A passion for excellence: The leadership difference. London: Collins.
- Peters, T. J., & Waterman, R. H. (1982). *In search of excellence: Lessons from America's best-run companies*. New York: Harper & Row.
- Philips, E. M., & Pugh, D. S. (2005). *How to get a PhD* (4th ed.). Glasgow: Open University Press.
- Porter, M. E. (1998). *Competitive advantage: Creating and sustaining superior performance*. New York, NY: Free Press.
- Puay, S. H., Tan, K. C., Xie, M., & Goh, T. N. (1998). A comparative study of nine national quality awards. *The TQM Journal, 10*(1), 30-39.
- QSR International. (2010). NVivo 9: Getting started. Retrieved 3 February, 2011, from http://download.qsrinternational.com/Document/NVivo9/NVivo9-Getting-Started-Guide.pdf
- Radziwill, N., Olson, D., Vollmar, A., Lippert, T., Mattis, T., Dewark, K. V., et al. (2008). Starting from scratch - Roadmap and toolkit: recipe for a new quality system. *Quality Progress*, *41*(9), 40-47.
- Ricondo, I., & Viles, E. (2005). Six Sigma and its link to TQM, BPR, Lean and the Learning Organisation. *International Journal of Six Sigma and Competitive Advantage*, *1*(3), 323-354.
- Rigby, D., & Bilodeau, B. (2005). The Bain 2005 management tool survey. *Strategy* & *Leadership*, 33(4), 4-12.
- Rigby, D., & Bilodeau, B. (2007). Bain's global 2007 management tools and trends survey. *Strategy & Leadership, 35*(5), 9-16.

- Rigby, D. K. (2007). *Management tools 2007: An executive's guide*. Boston, MA: Bain & Company Inc.
- Rimington, R. (2007). Managing Multiples Initiatives (MMI) / Integrated Management Systems (IMS) - A study on background and application. Singapore: Public Service Centre for Organisational Excellence.
- Ritchie, L., & Dale, B. G. (2000). Self-assessment using the business excellence model: A study of practice and process. *International Journal of Production Economics*, 66(3), 241-254.
- Sale, J. E. M., Lohfeld, L. H., & Brazil, K. (2002). Revisiting the quantitativequalitative debate: Implications for mixed-methods research. *Quality and Quantity, 36*(1), 43-53.
- Saunders, M. (2005). *Performance excellence and strategy deployment: A framework for implementing strategic initiatives.* Unpublished doctoral thesis, Massey University, Palmerston North.
- Saunders, M., Lewis, P., & Thornhill, A. (2009). *Research methods for business students* (5th ed.). Harlow: FT Prentice Hall.
- Saunders, M., & Mann, R. (2005). Self-assessment in a multi-organisational network. *International Journal of Quality & Reliability Management, 22*(6), 554-571.
- Saunders, M., & Mann, R. (2007). Business excellence tools: The tools used by companies at different stages of business excellence maturity. Retrieved 8 December, 2008, from http://www.coer.org.nz/ images/ files/ D136pub % 20Business % 20 Excellence% 20Tools%20Report.pdf
- Sawada, P. N. (1977). A concept of industrial engineering. *International Journal of Production Research*, *15*(6), 511-522.
- Seddon, J. (2003). *Freedom from command & control: A better way to make the work work*. Buckingham: Vanguard Education Ltd.
- Sharma, M., & Kodali, R. (2008). TQM implementation elements for manufacturing excellence. *The TQM Journal, 20*(6), 599-621.
- Sharp HealthCare. (2007). Malcolm Baldrige National Quality Award application report. San Diego: Sharp HealthCare.
- Shergold, K., & Reed, D. M. (1996). Striving for excellence: How self-assessment using the business excellence model can result in step improvements in all areas of business activities. *The TQM Journal, 8*(6), 48-52.
- Slack, N., Chambers, S., Johnston, R., & Betts, A. (2009). Operations and process management: Principles and practice for strategic impact (2nd ed.): Financial Times Prentice Hall.

- Slack, N., & Lewis, M. (2008). *Operations strategy* (2nd ed.). Harlow: Financial Times Prentice Hall.
- Snow, R. M., & Phillips, P. H. (2008). *Making critical decisions*. San Francisco: John Wiley & Sons Inc.
- Sousa, R., & Voss, C. A. (2008). Contingency research in operations management practices. *Journal of Operations Management, 26*, 697-713.
- Standards Australia / New Zealand. (2004). Handbook: Business Continuity Management - HB 221:2004 (2 ed.). Sydney and Wellington: Standards Australia / New Zealand.
- Standards Productivity and Innovation Board Singapore (SPRING). (2007). Award criteria. Retrieved 15 May, 2008, from http://www.spring.gov.sg/ Content/WebPage.aspx?id=0e082fc7-f1cf-464f-9857-3e692dfe69be
- Strauss, A., & Corbin, J. (1998). Basics of qualitative research: Techniques and procedures for developing grounded theory (2nd ed.). Thousand Oaks, CA: Sage Publications.
- Tan, K. C. (2002). A comparative study of 16 national quality awards. *The TQM Magazine, 14*(3), 165-171.
- Tan, K. C., Wong, M. F., Mehta, T., & Khoo, H. H. (2003). Factors affecting the development of national quality awards. *Measuring Business Excellence*, 7(3), 37-45.
- Tashakkori, A., & Teddlie, C. (1998). *Mixed methodology: combining the qualitative and quantitative approaches*. California: Sage publications.
- Tashakkori, A., & Teddlie, C. (2003a). Glossary. In A. Tashakkori & C. Teddlie (Eds.), Handbook of mixed methods in social & behavioral research (pp. 703-718). Thousand Oaks, CA: Sage Publications.
- Tashakkori, A., & Teddlie, C. (2003b). The past and future of mixed methods research from data triangulation to mixed model designs. In A. Tashakkori & C. Teddlie (Eds.), *Handbook of mixed methods in social & behavioral research* (pp. 671-701). California: Sage Publications.
- Tashakkori, A., & Teddlie, C. (2009). Integrating qualitative and quantitative approaches to research. In L. Bickman & D. J. Rog (Eds.), *The SAGE Handbook of Applied Social Research Methods* (pp. 283-317). Thousand Oaks: Sage publications.
- Teddlie, C., & Tashakkori, A. (2003). Major issues and controversies in the use of mixed methods in the social and behavioral sciences In A. Tashakkori & C.
 Teddlie (Eds.), *Handbook of mixed methods in social & behavioral research* (pp. 3-50). California: Sage Publications.

- Teddlie, C., & Tashakkori, A. (2009). Foundations of mixed methods research: Integrating quantitative and qualitative approaches in the social and behavioral sciences. Thousand Oaks, CA: Sage Publications.
- Thawesaengskulthai, N. (2007). Selecting quality management and improvement initiatives: Case studies of industries in Thailand (Doctoral thesis). Retrieved from http://etheses.nottingham.ac.uk/298/1/ Natcha%27s_thesis_V16_% 28 submit_etheses%29.pdf
- Thawesaengskulthai, N. (2010). An empirical framework for selecting quality management and improvement initiatives. *International Journal of Quality & Reliability Management, 27*(2), 156-172.
- Thawesaengskulthai, N., & Tannock, J. D. T. (2008a). A decision aid for selecting improvement methodologies. *International Journal of Production Research*, 46(23), 6721–6737.
- Thawesaengskulthai, N., & Tannock, J. D. T. (2008b). Pay-off selection criteria for quality and improvement initiatives. *International Journal of Quality & Reliability Management*, 25(4), 366-382.
- Triantaphyllou, E. (2000). *Multi-criteria decision making methods: A comparative study*. Dordrecht: Kluwer Academic Publishers.
- Van der Wiele, A., Van Iwaarden, J. D., Dale, B. G., & Williams, A. R. T. (2007).
 Improvement approaches. In B. G. Dale, A. Van der Weile & J. Van Iwaarden (Eds.), *Managing Quality* (5th ed., pp. 559-575). Malden, MA: Blackwell Publishing.
- Van der Wiele, A., Williams, A. R. T., & Dale, B. G. (2000). ISO 9000 series registration to business excellence: The migratory path. *Business Process Management Journal, 6*(5), 417-427.
- Voss, C. A. (2005). Paradigms of manufacturing strategy re-visited. *International Journal of Operations & Production Management, 25*(12), 1223-1227.
- Walliman, N. (2006). Social research methods. London: SAGE Publications.
- Weiler, G. (2004). What do CEOs think about quality? *Quality Progress, 37*(5), 52-56.
- Wieleman, B. P. (2011). Selecting business improvement methods: Towards a technique for consultants to support the selection of methods in an improvement project. Unpublished master thesis, Eindhoven University of Technology, Eindhoven, Netherlands.

- Williams, R., Bertsch, B., van der Wiele, A., Van Iwaarden, J., & Dale, B. G. (2006). Self-assessment against business excellence models: A critique and perspective. *Total Quality Management & Business Excellence, 17*(10), 1287-1300.
- Wilson, D. D., & Collier, D. A. (2000). An empirical investigation of the Malcolm Baldrige National Quality Award causal model. *Decision Sciences*, 31(2), 361-383.
- Yang, C. C. (2004). An integrated model of TQM and GE-Six-Sigma. *International Journal of Six Sigma and Competitive Advantage, 1*(1), 97-111.
- Yoon, K. P., & Hwang, C. L. (1995). *Multiple attribute decision making: An introduction*. California: Sage Publication Inc.
- Zhang, Z. (2000). Developing a model of quality management methods and evaluating their effects on business performance. *Total Quality Management* & *Business Excellence*, 11(1), 129-137
- Zink, K. J., & Steimle, U. (2000). Corporate social responsibility and organizational excellence Linking the Stakeholder approach and Resource-Based Views.
 In K. J. Foley & P. Hermel (Eds.), *The Theories and Practices of Organizational Excellence: New Perspectives* (pp. 147-176). Sydney: SAI Global.
- Zutshi, A., & Sohal, A. S. (2005). Integrated management system: The experiences of three Australian organisations. *Journal of Manufacturing Technology Management, 16*(2), 211-232.